

# Adoption of IFRS in Spain: Effect on the comparability and relevance of financial reporting

Susana Callao\*, José I. Jarne, José A. Laínez

Universidad de Zaragoza, Facultad de CCEE y EE, Departamento de Contabilidad y Finanzas, Gran Vía 2, 50005 Zaragoza, Spain

#### Abstract

EU Regulation 1606/2002 requires application of International Financial Reporting Standards (IFRS) by groups listed on European stock markets. In Spain, listed groups are now obliged to prepare consolidated financial information under IFRS, and legislative changes to bring local rules into line with international standards have been tabled.

In this context, the potential impact of IFRS is fraught with uncertainty. Our study of IBEX-35 companies focuses on the effects of the new standards on comparability and the relevance of financial reporting in Spain. We address these objectives by seeking significant differences between accounting figures and financial ratios under the two sets of standards (i.e. Spanish accounting standards and IFRS).

The results obtained show that local comparability has worsened. The study reveals that local comparability is adversely affected if both IFRS and local accounting standards are applied in the same country at the same time. Reforms to bring local rules into line with international standards are therefore urgent. We also find that there has been no improvement in the relevance of financial reporting to local stock market operators because the gap between book and market values is wider when IFRS are applied. While there has been no gain in terms of the usefulness of financial reporting in the short-term, improved usefulness may be achieved in the medium to long-term.

© 2007 Elsevier Inc. All rights reserved.

Keywords: International Financial Reporting Standards (IFRS); Financial ratios; Financial reporting usefulness; Spanish stock market

### 1. Introduction

The emergence and development of multinational concerns, the growth of international financial markets and changing investor behavior have, among other factors, contributed to the

\* Corresponding author. Tel.: +34 976 761791; fax: +34 976 761769. *E-mail address:* scallao@unizar.es (S. Callao).

<sup>1061-9518/\$ -</sup> see front matter © 2007 Elsevier Inc. All rights reserved. doi:10.1016/j.intaccaudtax.2007.06.002

internationalization of economic activity. As a result of this phenomenon, financial reporting has spread beyond national borders.

However, interpretation and understanding of financial information at the international level is hindered by a multitude of factors, including the diversity of the accounting principles and rules governing the preparation of reports.

Considerable efforts have been made by various bodies (International Accounting Standards Board (IASB) and the European Union (EU)), since the 1970s to harmonize accounting rules in different countries, with the aim of improving the usefulness of financial information in the international context. Among other factors, the non-mandatory nature of the standards issued by the IASB, the flexibility and ease of compliance with EU Directives and, most seriously, the lack of political will in the countries concerned, rooted in local culture and a strong national outlook, have so far prevented the attainment of a truly harmonized framework for useful financial reporting.

Awareness within the EU of the need to make progress towards achieving international comparability resulted in the approval of Regulation 1606/2002, which provides for the mandatory application of International Financial Reporting Standards (IFRS) by business groups listed on European stock markets as of January 2005. The approval of this regulation has resulted in the adoption of IFRS in European countries. This paper focuses on the adaptation process in Spain.

#### 1.1. Spanish accounting system

In contrast to those countries where professional practice orients accounting standards, the accounting rules in Spain, a country where the legal system is based on Roman Law, are enshrined in legislation. Hence, accounting rules in Spain have traditionally been a public issue and there has been scant input from the private sector. Regulations have taken the form of companies legislation (e.g. the Commercial Code and the Spanish Companies Act), the General Chart of Accounts and the related implementing Regulations, and other Securities Market and Bank of Spain legislation. Consequently, the Spanish accounting system is of a Continental European nature.

The adoption of IFRS in Spain was implemented by the Ministry of Economy with the publication of Law 62/2003, 30 December, establishing tax, administrative and social measures. This Act provides for the application of the following accounting rules, applicable exclusively to consolidated financial statements prepared by companies required to report mandatory consolidated accounting information in years commencing as of January 1, 2005:

- (a) If at the closing date for the year, any group company may have issued securities listed in an official market in any member State of the EU, they shall apply the IFRS approved in the Regulations of the European Commission<sup>1</sup>.
- (b) If at the closing date for the year, no group company may have issued securities listed in an official market in any member State of the EU, they may opt to apply either Spanish accounting standards or the IFRS approved in the Regulations of the European Commission. If they opt for the latter, the consolidated financial statements shall be prepared in accordance with such standards on an ongoing basis.

<sup>&</sup>lt;sup>1</sup> Nevertheless, companies (except financial institutions) required to prepare consolidated financial statements but that may have issued only *fixed interest securities* listed on an official market of any member State of the EU at the close of 2005, may continue to apply Spanish accounting standards until years commencing as from January 1, 2007, unless they have already applied the IFRS approved by the Regulations of the European Commission in any prior year.

The regulatory change has raised numerous questions concerning the potential effects of the application of IFRS in an accounting environment that is unaccustomed to the utilization of accounting rules based on the structure and philosophy underlying international standards. Will the desired comparability of financial information at the European level be achieved if IFRS are applied only to consolidated financial statements? What will happen to individual financial statements or to the consolidated accounts of listed and unlisted groups? Will the local comparability of financial information improve? What will be the effects of the change on the stock market? Will there be any improvement in the value relevance of listed firms' financial reporting?

In our opinion, the key issue is whether the application of IFRS will improve the usefulness of financial reporting. To be useful, according to IASB Framework, information must be understandable, reliable, comparable and relevant. No doubt, financial statements prepared under IFRS will be more useful when they are used in an international context. But we are concerned about the effects of IFRS on the usefulness of financial reporting in the national context. This is the issue examined in this paper.

This research focuses on the effects of IFRS on the comparability and relevance of financial reporting in Spain. These are two of the four key qualitative characteristics that make financial information useful.

We seek to establish whether the financial statements of Spanish firms are comparable when some firms apply IFRS and others apply local standards. For this purpose, we measure the quantitative impact of the new rules for recognition and valuation on financial figures and ratios. The *a priori* effect of IFRS should be significant, to the extent that they contain criteria that are clearly different from Spanish accounting standards. For example, Laínez, Callao, and Jarne (2004) describe the expected effects of the first application of international standards on balance sheet captions. In any event, it is necessary to remember that the quantitative impact will depend on the accounting policies adopted by the Spanish firms themselves since IFRS provide a wide range of options.

Financial reporting is relevant when it influences the economic decisions of users, whether investors, employees, lenders, suppliers, customers or other agents. We focus on the investors and analyze whether IFRS make financial reporting more relevant for decision making in the capital markets than the information provided in financial statements prepared under Spanish standards. To this end, we analyze the effect on the gap between firms' book and market values and, by extension, the impact on the book-to-market ratio.

We expect our results to be of great interest to academics involved in guiding and researching progress with international accounting harmonization, and to the Spanish regulatory and supervisory authorities (Institute of Accounting and Auditing, National Securities Market Commission and the Bank of Spain) since the study provides insight into the results of implementing IFRS.

Our research may also be relevant to international regulators and institutions involved in the process (e.g. the European Commission, EFRAG, IASB and the securities markets), since the results provide examples of how firms required to apply IFRS have approached the process in a continental European accounting system characterized by regulatory rigidity and a legalistic outlook.

The remainder of the paper is organized as follows. The next section presents previous literature in the field. Section 3 describes the main differences between Spanish rules and IFRS. Details of the sample, data and methodology are given in Section 4. Section 5 deals with the impact of IFRS on comparability. Section 6 describes a qualitative analysis of the recognition and valuation policies that have generated the changes observed. Section 7 addresses the second objective, which concerns the impact of IFRS on the relevance of financial information. Finally, we discuss the main conclusions.

#### 2. Literature review

A number of studies related to the objectives of this paper have been published in recent years, which we shall consider in two groups. The first comprises studies analysing the impact of the EU's adoption of IFRS on different aspects of national accounting models, while the second refers to papers examining the effect on the value relevance of the accounting information produced under different sets of standards.

Since the official announcement of the mandatory application of IFRS in the European Union in 2002, accounting research has examined the process, especially the impact, progress and difficulties it entails. Table 1 presents an overview of prior research on implanting IFRS.

Jermakowicz (2004) analyzes the IFRS adoption process in Belgium, which, like Spain, is an example of the continental accounting model. The paper analyzes the impact of IFRS on BEL-20 firms. A survey sent to Belgian companies indicates that implementing IFRS will dramatically change the way these companies design and handle both their internal and external reporting activities, and will increase the comparability of consolidated accounts as well as levels of transparency for many companies. The quantitative impact is only analyzed in three companies, which were the first companies to adopt IFRS in 2003. The study concludes that adjustments to translate Belgian GAAP to IFRS resulted in a significant impact on the companies' reported equity, as well as net income.

Ormrod and Taylor (2004) study the impact of the change from UK GAAP to IFRS on covenants included in debt contracts. They suggest that the change is likely to result in more volatile reported earnings figures, in addition to differences in reported profits and balance sheet items. A movement towards cash flow-based covenants might thus be seen as one method of moderating the uncertainty for borrowers arising from the introduction of IFRS.

Weißenberger, Stahl, and Vorstius (2004) survey the motives that led certain German companies to opt for US GAAP or IFRS rather than German GAAP. The authors sent a questionnaire to a sample of 359 companies (DAX100 and *Neuer Markt*) and received 81 responses. Their results indicate that the change to IFRS or US GAAP was motivated by the expectation of gaining standing in the capital markets, achievement of improved supply of information, and the internationalization of investors. However, an ex-post evaluation revealed that not all of these objectives were achieved.

Larson and Street (2004) examine progress towards and perceived impediments to convergence in 17 European countries (the 10 new EU members, Switzerland and other EU candidate countries). The authors employ data collected by the former Big Six international accounting firms in their 2002 convergence survey. The results suggest the emergence of a "two-standard" system (IFRS for consolidated financial statements of listed companies and local GAAP for non-listed companies). The two most significant impediments to convergence are the complexity of certain IFRS and the tax-orientation of many national systems.

Schipper (2005) describes a series of implementation effects associated with the mandatory adoption of IFRS in the EU. The IASB has found it necessary to provide detailed implementation guidance for IFRS, otherwise preparers and auditors turn to US GAAP or jurisdiction-specific European GAAP. Likewise, the adoption of IFRS coupled with the IASB's commitment to international convergence with the FASB will place additional pressure on two reporting issues: defining the reporting entity for consolidation purposes and developing reliable fair value measures.

Van Tendeloo and Vanstraelen (2005) address the question of whether the adoption of IFRS is associated with lower earnings management. They examine whether German companies that have adopted IFRS engage significantly less in earnings management compared to firms reporting

### Table 1 Literature on adoption of IFRS in the EU

References	Country analyzed	Торіс	Results
Jermakowicz (2004)	Belgium	Impact of conversion to IFRS on companies, their internal organization and accounting and finance strategy	Significant changes in internal and external reporting activities, and impact on their reported equity and net income
Ormrod and Taylor (2004)	United Kingdom	Impact of the change to IFRS on covenants in companies' debt contracts	Changes in reported profits and balance sheet amounts, and more volatile reported earnings figures
Weißenberger et al. (2004)	Germany	Motives that led German companies to opt for US GAAP or IFRS rather than German GAAP	Expectation of attaining improved standing in the capital markets
Larson and Street (2004)	17 European countries	Progress and impediments to convergence to IFRS	Development of "two-standard" system (IFRS for consolidated financial statements of listed companies and local GAAP for non-listed companies); the two most significant impediments are the complicated nature of certain IFRS and the tax-orientation of many national systems
Schipper (2005)	_	Implementation effects associated with the mandatory adoption of IFRS	IASB needs to provide detailed implementation guidance for IFRS
Van Tendeloo and Vanstraelen (2005)	Germany	Effects of IFRS on earnings management	Adoption of IFRS cannot be associated with lower earnings management
Tokar, 2005	-	Impact of convergence to IFRS on auditing firms	Investment of money and a significant change in the training of accounting studies
Delvaille et al. (2005)	France, Germany and Italy	Comparison of developments to integrate IFRS in the three countries	Use of IFRS is most prominent in Germany, and there are significant differences in the way the three countries adapt to the IFRS
Jones and Luther (2005)	Germany	Consequential effects of the change to IFRS on management accounting practices	Managers need to choose between integrating external and internal reporting or operating dual accounting systems
Callao et al. (2006)	Spain	Approach to the process of implementing IFRS in Spanish firms and implications	The cost of adaptation is high for firms and they foresee major changes in recognition and measurement policies

under German GAAP. Their results suggest that the adoption of IFRS cannot be associated with lower earnings management.

Tokar (2005) focuses on the impact of convergence on auditing firms and concludes that achieving true convergence of accounting standards is a costly and time-consuming objective, and will require a huge investment of money and a significant change in the training of accounting students in the near future.

Delvaille, Ebbers, and Saccon (2005) compare developments in France, Germany and Italy and the approaches taken to integrate the current European accounting reform processes with IFRS. They make an empirical analysis of the use of IFRS by listed companies on three stock markets (DAX 30, CAC 40 and S&P/MIB). The results show clearly that the use of IFRS is most prominent in Germany. They conclude that while France, Germany and Italy were in the past considered to employ Continental European accounting systems, they are very different today, not only in reporting practice but also in the way they have adapted to IFRS.

Jones and Luther (2005) examine whether the change to IFRS in Germany could have consequential effects on the distinctive traditional management accounting practices applied in the field of control. They examine three Bavarian companies and two management consultancy firms, concluding that managers face an important choice between integrating external and internal reporting or continuing to operate dual accounting systems, confining the adoption of IFRS to external reporting. At present no trend has been established.

Callao, Jarne, and Laínez (2006) investigate the manner in which Spanish firms have handled the process of implementing IFRS. The empirical work is based on a survey of Spanish business groups listed on the Madrid stock exchange. The results show that listed Spanish groups have taken a very positive stance towards the harmonization process and the adoption of IFRS, but adaptation is costly and requires changes in business organization and structures, as well as accounting policies.

The business and other issues analyzed in connection with the adoption of IFRS in Europe are very varied, although few studies have addressed the impact of adoption on accounting figures. In fact, our study is one of the first to consider these issues in the field of European accounting information.

In the second group of papers, referring to the effects of legislative change on the value relevance of accounting information, there are still very few published studies concerning the actual process of IFRS adoption in Europe. The papers which are briefly summarized in Table 2, are related to the adoption of IFRS at different times and countries.<sup>2</sup>

Harris and Muller (1999) examine the market valuation of earnings and book value figures prepared under IFRS and US-GAAP.<sup>3</sup> The sample consists of 89 non-US companies that employ IFRS in their primary accounts with a reconciliation to US GAAP during the period 1992–1996, and the models used are based on earnings and Ohlson models. They find evidence that US GAAP earnings reconciliation amounts are value-relevant after controlling for IFRS amounts in market value and return models. IFRS amounts are more (less) closely associated with prices-per-share (security returns) than US GAAP amounts.

<sup>&</sup>lt;sup>2</sup> There are other articles that analyze the value relevance of accounting figures produced under local GAAP and US GAAP. For example Alford, Jones, Leftwich, and Zmijewski (1993), Amir, Harris, and Venuti (1993), Bandyopadhyay, Hanna, and Richardson (1994), Harris, Lang, and Möller (1994), Chan and Seow (1996) or Mora and Rodríguez (2004).

<sup>&</sup>lt;sup>3</sup> Though some of these papers refer to IASB rules as International Accounting Standards (IAS), we have opted here to use the current terminology, referring to International Financial Reporting Standards (IFRS).

 Table 2

 Literature on the effects of IFRS on financial information relevance

References	Sample	Period	Model	Results
Harris and Muller (1999)	31 companies (reconciliation IFRS-US GAAP)	1992–1996	Multiple linear regression (Earnings and Ohlson models)	Reconciliations are value-relevant, IFRS are more closely associated with prices-per-share than US GAAP but US GAAP are more closely associated with returns than IFRS
Niskanen et al. (2000)	18 companies (reconciliation Finnish GAAP-IFRS)	1984–1992	Multiple linear regression (Earnings model)	Reconciliations do not appear to be value-relevant
Bartov et al. (2005)	417 companies (US GAAP, German GAAP and IFRS)	1998–2000	Linear regression (cross sectional and time series)	US GAAP and IFRS are more value relevant than German GAAP
Lin and Chen (2005)	415 companies (reconciliation Chinese GAAP-IFRS)	1995–2000	Multiple linear regression (Earnings and Ohlson models)	Chinese GAAP more value relevant than IFRS
Horton and Serafeim (2006)	85 companies (reconciliation UK GAAP-IFRS)	2005	Multiple linear regression (event and value relevance studies)	Reconciliation adjustments in respect of earnings (but not in respect of shareholders' equity) are value relevant
Schiebel (2006)	12 German GAAP 12 IFRS	2000–2004	Linear and exponential regression (panel data)	German GAAP more value relevant than IFRS

The value relevance of earnings under Finnish accounting standards and their reconciliations to IFRS are studied by Niskanen, Kinnunen, and Kasanen (2000). They analyzed 18 Finnish firms that disclose earnings under local GAAP and IFRS (1984–1992) using an earnings model. The results show that the change in local GAAP earnings, as well as the level and change in aggregate reconciliation to IFRS, are value irrelevant.

Bartov, Goldberg, and Kim (2005) investigate the comparative value relevance of earnings reported under German GAAP, US GAAP and IFRS. The research sample included 417 German companies listed on local stock markets during the period 1998–2000. They use a return model and conclude that US GAAP and IFRS are more value relevant than German GAAP.

Outside the EU, Lin and Chen (2005) investigate the incremental value relevance obtained from reconciling accounts prepared under Chinese accounting standards to IFRS. The sample comprises between 53 and 79 companies per year listed on Chinese stock markets for the period 1995–2000. The authors apply the Ohlson model and the returns model and find that earnings and the book values of equity determined under Chinese GAAP provide more relevant accounting information for the purpose of determining the prices of shares than IFRS.

Addressing the adoption of IFRS in EU, Horton and Serafeim (2006) examine the market reaction to and value relevance of reconciliation adjustments from UK companies in the transition to IFRS compliance. The sample consists of 85 firms from the London Stock Exchange FTSE 350 for 2005. The authors employ an event study methodology and a market value model. They find the reconciliation adjustment from UK GAAP to IFRS to be value relevant with respect to earnings but not to shareholders' equity.

In Germany, Schiebel (2006) examines the value relevance of IFRS and German GAAP. The sample include 24 German companies listed on the Frankfurt stock exchange (12 companies publishing exclusively German GAAP consolidated reports for the period 2000–2004 and 12 companies publishing exclusively IFRS consolidated reports for the period 2000–2004). The author proposes different regressions of market capitalization on consolidated equity book value using a simple linear regression analysis, finding that German GAAP are significantly more value relevant than IFRS.

These findings are very mixed, with some studies showing that the change to IFRS improves value relevance (Bartov et al., 2005; Harris & Muller, 1999; Horton & Serafeim, 2006), and others that it worsens value relevance (Lin & Chen, 2005; Schiebel, 2006), while yet others find no conclusive evidence either way (Niskanen et al., 2000).

The second objective of our study is closely linked to the last two papers referred to above. We examine the improvement in the value relevance of accounting information as a result of the application of IFRS rather than local criteria.

#### 3. Key differences between Spanish accounting standards and IFRS

Until 2005 all Spanish firms prepared their financial statements in accordance with the local standards issued in 1990 following EU Directives (Fourth Council Directive 78/660/EEC of 25 July 1978, on the annual accounts of certain types of companies, and Seventh Council Directive 83/349/EEC of 13 June 1983 on consolidated accounts). For the companies in this study, the applicable rules are in the Spanish General Chart of Accounts (1990) and the Regulations for the Preparation of Consolidated Annual Accounts (1991).

As in are EU member States, listed Spanish groups are now required to adopt IFRS, which means adapting their accounting policies to international standards. Accounting issues that are likely to generate the greatest differences with the rules applied to date are discussed below

and will help to understand the reasons for the changes in the different areas of financial information.

Appendix A describes the principal differences between Spanish Accounting Standards (SAS) and IFRS. We have classified the differences observed in three groups, depending on whether the issue is mainly related to the preparation of the balance sheet (balance sheet areas), or to the calculation of income (income areas). The third group includes issues that cannot be easily classified as pertaining to the balance sheet or income. The table has four columns. The first defines the issue, the second and third explain the IFRS and SAS treatment, respectively, and the fourth describes the differences between IFRS and SAS.

As may be observed, differences appear in many balance sheet areas, including tangible and intangible assets, inventories, grants, goodwill and provisions. There are also significant differences in areas related to the calculation of income such as start-up costs, research and development expenditure, borrowing costs, leases, extraordinary results and income taxes. Finally, significant differences are also found in other areas, such as foreign currency, investment in associates, changes in accounting policies and errors.

It should not be forgotten that Spanish standards are only minimally developed, or even nonexistent, in a number of areas such as investment property, non-current assets held for sale, financial instruments, employee benefits and financial reporting in hyperinflationary economies.

In some cases, such as valuation of property, plant and equipment, valuation of intangible assets or valuation of financial instruments, IFRS allow more than one option for accounting treatments, and one or more of the permitted criteria are already established in Spanish standards. Consequently, the differences found at the regulatory level do not always take shape in accounting practice. Where IFRS provide options, the differences in accounting standards take shape in accounting practice depending on whether firms opt for accounting policies that coincide with those already applied in accordance with local standards, or for divergent treatments. For example, IAS 16 permits the valuation of fixed assets at acquisition cost (a mandatory criterion under Spanish standards) or at revalued cost. This matter, which results in differences between the two standards, need not result in differences in practice if Spanish firms opt to continue valuing their assets at acquisition cost.

Consequently, this study examines the actual impact of IFRS on reporting by Spanish firms since it is likely that not all differences between standards will have produced variations in the financial statements prepared in the transition from one set of standards to the other.<sup>4</sup>

#### 4. Sample and methodology

The sample comprises IBEX 35 firms at June 30, 2005. These are the firms with the highest stock market capitalization on the continuous market and, therefore, they are representative of the behavior and evolution of the Spanish stock market over a given period.

We have eliminated financial institutions and insurance companies from the total IBEX 35 firms at the aforementioned date in view of the peculiarities of the financial industry and its specific regulation. One firm was also discarded, because it had applied IFRS since 2002. As a result, the sample comprises 26 firms.

<sup>&</sup>lt;sup>4</sup> Finally, Spanish accounting standards (applicable to unlisted groups and stand-alone firms) are currently in the process of adaptation to international standards. It is expected that this process will be completed in 2007 with the publication of a new General Chart of Accounts that is more closely in line with international trends.

The study is based on the 6-monthly information reported by these firms to the Spanish National Securities Market Commission in the first half of 2004 and 2005. The information referring to the first half of 2004 was prepared under Spanish accounting standards. The data for the first half of 2004 under international standards (presented as comparative figures) and the reconciliation of the closing balance sheet for 2004 under international and Spanish standards has been extracted from the interim information for 2005.

The Spanish National Securities Market Commission published the Circular 1/2005, April 1, in order to regulate the transition from national to international standards. This Circular amends the rules for reporting by firms with securities traded on the Spanish stock market.

In accordance with Circular 1/2005, listed Spanish groups required to apply IFRS since January 2005 were also obliged to file interim information for the first half of the year using specific formats which include, the following:

- The consolidated opening balance sheet for the business year commencing in 2005 prepared under IFRS together with the closing balance sheet for the year ended in 2004 prepared in accordance with the local standards applicable in that year.
- The balance sheet and income statement for the first half of 2005 with comparative figures for the prior year. This information must be prepared in accordance with the recognition and valuation principles applied by the group to prepare its first set of consolidated financial statements in accordance with the IFRS adopted.
- A detailed explanation of significant adjustments made in order to evaluate the effect of applying all of the IFRS adopted on the opening balance sheet for the year commencing in 2005.

These disclosure requirements allow us to consider both the balance sheet and the income statement for the first half of 2004 prepared under international rules (information included in the comparative figures disclosed together with the first half of 2005) and under Spanish accounting standards (information disclosed for the first half of 2004). We also have the closing balance sheet for 2004 prepared under Spanish standards and the opening balance sheet for 2005 under IFRS.

We address the objectives mentioned by seeking the existence of significant differences between variables, which are compared for the same subjects or firms under different conditions (related samples). The study is performed applying parametric and non-parametric tests depending on whether or not the variables concerned follow a normal distribution.

To test normality we use the Kolmogorov–Smirnov (with Liliierfors significance correction) and Shapiro–Wilks tests (Appendix B). The *t*-test for related samples is then applied to the variables found to be normal. Non-parametric tests (specifically the Wilcoxon signed-ranks test) are applied to non-normal variables.

#### 5. Impact of International standards on financial reporting comparability

The objective of the study is to test for the presence of significant differences in key figures and ratios calculated on the basis of Spanish and international accounting standards, respectively denoted SAS and IFRS. The comparison was performed on interim information for the first half of 2004 because this is the first period for which figures for the balance sheet and income statement are available under both Spanish and international rules, as explained above. We then examine whether the results obtained persist in the 2004 year-end information.

### 5.1. Definition of variables and hypothesis

The interim information is analyzed on the basis of the following variables:

 $F_{iSAShalf}$  value of the variable  $F_i$  under SAS at the close of the first half of 2004  $F_{iIFRShalf}$  value of the variable  $F_i$  under IFRS at the close of the first half of 2004

where F represents the following figures and ratios:

- balance sheet figures (fixed assets, inventories, debtors, cash, current assets, total assets, equity, long-term liabilities, short-term liabilities, total liabilities, long-term resources, total equity and liabilities);
- income statement lines (operating income, ordinary income, net income and net income attributable to equity holders of the parent);
- financial ratios (current ratio, acid test, cash ratio, solvency, indebtedness, return on assets per operating income and ordinary income, return on equity per ordinary income and net income).

Table 3 presents the definitions of these acconting figures and financial ratios.

In total, the data base comprises 25 variables measured under two sets of rules, local and international standards. Descriptive statistics for the variables are included in Appendix C (Panel 1). The null hypothesis tested for each of these variables is:

 $H_{01}$ . There are no significant differences in the value taken by the variable  $F_i$  in the interim information under SAS and IFRS.

The variables for the data base of annual information (descriptive statistics are included in Appendix C (Panel 2)) are the same with the exception of those referring to results (operating income, ordinary income, net income and net income attributable to equity holders of the parent, return on assets and return on equity), since only the balance sheet per Spanish and international standards is available, but not the income statement. In this case, 17 figures and ratios are measured under the two sets of rules:

 $F_{iSASyear}$  value of the variable  $F_i$  under SAS at the close of 2004.  $F_{iIFRSyear}$  value of the variable  $F_i$  under IFRS at the close of 2004.

In this case, the null hypothesis tested is:

**H**<sub>02</sub>. There are no significant differences in the value taken by the variable  $F_i$  in the annual information under SAS and IFRS.

### 5.2. Results

#### 5.2.1. Differences in interim information

The null hypothesis  $H_{01}$  was analyzed using parametric and non-parametric tests (respectively, the *t*-test and the Wilcoxon signed-ranks test) depending on the normality or otherwise of the variables. The results are presented in Table 4.

 Table 3

 Accounting figures and financial ratios

Figures	Definition
Fixed assets	Intangible assets + property, plant and equipment + long-term investments + goodwill
Inventories	Goods produced for sale + goods in the process of production + materials or supplies
Debtors	Receivables + short-term investments
Cash	Cash + cash equivalents
Current assets	Inventories + debtors + Cash (as defined above)
Total assets	Fixed assets + current assets (as defined above)
Equity	Funds contributed by shareholders + retained earnings + other reserves + net income + minority interest + deferred income
Long-term liabilities	Long-term creditors + long-term provisions
Short-term liabilities	Short-term creditors + short-term provisions
Total liabilities	Long-term liabilities + short-term liabilities (as defined above)
Long-term resources	Equity + long-term liabilities (as defined above)
Equity + liabilities	Equity + total liabilities (as defined above)
OPI (Operating income)	Operating income – Operating expenses
ORDI (Ordinary income)	OPI+Financial income – Financial expenses
NETI (Net income)	ORDI+extraordinary income - extraordinary expenses - taxes
ATTI (Net income attributable to the parent)	NETI – net income attributable to minority interest
Ratios	Definition
Current ratio	Current assets/short-term liabilities (as defined above)
Acid test	(Debtors + cash)/short-term liabilities (as defined above)
Cash ratio	Cash/short-term liabilities (as defined above)
Solvency	Total assets/total liabilities (as defined above)
Indebtedness	Total liabilities/equity (as defined above)
ROA (OPI)	Operating income/total assets (as defined above)
ROA (ORDI)	Ordinary income/total assets (as defined above)
ROE (ORDI)	Ordinary income/equity (as defined above)
ROE (NETI)	Net income/equity (as defined above)

The null hypothesis was rejected for twelve 12 at a maximum error level of 10%. Hence, five balance sheet items display significant differences (at 1%) depending on whether Spanish or international standards are applied. These are debtors, cash, equity, long-term liabilities and total liabilities. In the income statement, significant differences (at 5%) were observed at the level of operating income.

Table 4 Test of hypothesis  $H_{01}$  (observations related to 26 firms)

	Statistic <sup>b</sup>	Sig.
Figures		
Fixed assets <sub>half</sub>	-0.647	0.517
Inventories <sub>half</sub>	-0.845	0.398
Debtorshalf	-2.882	0.003*
Cash <sub>half</sub>	-3.658	0.000*
Current assetS <sub>half</sub>	-0.723	0.469
Total assets <sub>half</sub>	-0.114	0.909
Equity <sub>half</sub>	-3.263	0.001*
L.T. liabilities <sub>half</sub>	-3.441	0.000*
S.T. liabilities <sub>half</sub>	-0.174	0.861
Total liabilitieshalf	-3.517	0.000*
L.T. resourceShalf	-0.774	0.438
Equity + liabilitieshalf	-0.114	0.909
OPI <sub>half</sub>	-2.527	0.011**
ORDI <sub>half</sub>	-0.063	0.949
NETI <sub>half</sub>	-0.038	0.969
ATTI <sub>half</sub>	-0.571	0.567
Ratios		
Current ratio <sub>half</sub> <sup>a</sup>	0.726	0.475
Acid test <sub>half</sub> <sup>a</sup>	1.455	0.158
Cash ratio <sub>half</sub>	-3.619	0.000*
Solvency <sub>half</sub>	-3.593	0.000*
Indebtedness <sub>half</sub>	-2.933	0.003*
ROA (OPI) <sub>half</sub>	-2.019	0.043**
ROA (ORDI) <sub>half</sub>	-0.088	0.929
ROE (ORDI)half	-2.501	0.012**
ROE (NETI) <sub>half</sub>	-2.171	0.029**

\*Significant at 1%; \*\*significant at 5%.

<sup>a</sup> Normal variable.

<sup>b</sup> Statistic Z (Wilcoxon test) for no normal variables and t statistic for normal variables.

Finally, six financial ratios reflected the presence of significant differences due to the application of the different standards. These were cash ratio, solvency and indebtedness (at 1%), return on assets per operating income, return on equity per ordinary income and net income (at 5%).

Based on the number of positive and negative ranks, as well as the sum of ranks of each sign provided by the Wilcoxon test, we may determine the sign of the variations experienced by the different variables. To focus on the variables generating significant variations as a result of the change from Spanish to international standards, we conclude that the financial statements of Spanish firms adopting IFRS reflect:

- Increases in cash and cash equivalents, long-term and total liabilities and in the cash ratio, indebtedness and return on equity.
- Decreases in debtors, equity, operating income and the solvency ratio and return on assets (measured in terms of the operating income).

Table 5 Test of hypothesis  $H_{02}$  (observations related to 26 firms)

	Statistic <sup>b</sup>	Sig.
Figures		
Fixed assets <sub>year</sub>	-1.257	0.208
Inventories <sub>vear</sub>	-0.149	0.881
Debtorsyear	-3.416	0.000*
Cash <sub>year</sub>	-3.290	0.001*
Current assets <sub>year</sub>	-2.400	0.016**
Total assets <sub>year</sub>	-0.012	0.989
Equity <sub>year</sub>	-2.603	0.009*
L.T. liabilities <sub>year</sub>	-2.958	0.003*
S.T. liabilities <sub>year</sub>	-0.876	0.380
Total liabilities <sub>year</sub>	-2.958	0.003*
L.T. resources <sub>year</sub>	-0.850	0.394
Equity + liabilities <sub>year</sub>	-0.012	0.989
Ratios		
Current ratiovear <sup>a</sup>	1.050	0.304
Acid test <sub>vear</sub>	-2.908	0.003*
Cash ratio <sub>year</sub>	-3.416	0.000*
Solvency <sub>year</sub>	-2.273	0.023**
Indebtedness <sub>year</sub>	-1.968	0.049**

\*Significant at 1%; \*\*significant at 5%.

<sup>a</sup> Normal variable.

<sup>b</sup> Statistic Z (Wilcoxon test) for no normal variables and *t* statistic for normal variables.

#### 5.2.2. Differences in annual information

This section describes the results obtained from testing whether the variables undergoing significant variations in the interim information for 2004 are the same in the year-end financial statements.

As noted in Table 5, the variables for which significant differences are found are practically the same as in the interim information, but they are joined by current assets (at 5%) and the acid test ratio (at 1%).

The sign of the differences observed for the significant differences in the interim information is the same in all cases in the analysis of annual data. The two variables that were found not to be significant in the interim information, current assets and the acid test ratio, decrease due to the application of IFRS.

We conclude from the initial analysis on the variables exhibiting significant differences that the results obtained for the interim information are consistent with those found in the annual information scenario.

#### 6. Main issues underlying differences in financial information

In this section, we analyze the issues underlying the adjustments required to adapt the Spanish firms' financial information to IFRS. The nature of the information disclosed by the sample firms with regard to the adjustments does not allow statistical analysis, and three of them in fact failed to provide any explanatory notes. Consequently, the following discussion is based on a total of 23 firms.

Numerous issues have resulted in adjustments for Spanish firms. However, certain adjustments were made by more than 50% of the firms in the sample. We focus on the issues producing these adjustments, which are goodwill amortization, measurement of financial instruments, recognition of research costs, recognition of extraordinary income and expenses, accounting method for deferred tax and recognition of deferred tax assets.

Under Spanish GAAP, goodwill is amortized over the useful life of the investment, which may not be exceed 20 years. According to IFRS 3 goodwill acquired in a business combination shall not be amortized, instead the acquirer shall test it for impairment annually.

In accordance with Spanish standards, financial assets are measured at the lower of acquisition cost and market value, while financial liabilities are measured at repayment value. IAS 39 recommends that after initial recognition, financial assets be measured at their fair values, except loans and receivables, held-to-maturity investments and investments in equity instruments that do not have a quoted market price and whose fair value cannot be reliably measured. Financial liabilities should be measured at amortized cost using the effective interest method<sup>5</sup>.

In Spain, R&D expenditure may be recognized as an intangible asset if certain conditions are found. Under IAS 38, research expenditure shall be recognized as expense when they are incurred and intangible assets do not arise. The development expenditures should be recognized as an intangible asset when certain conditions can be found.

The income statement in Spain uses a classification of expenses based on their nature and distinguishes between ordinary and extraordinary income and expenses. Under IAS 1, an entity shall not present any items of income and expense as extraordinary items. Spanish firms have allocated the extraordinary items to the rest of ordinary items according to their nature.

Spanish GAAP require an entity to account for deferred tax using the income statement liability method, while IAS 12 requires the balance sheet liability method. The former focuses on timing differences (i.e. differences between taxable and accounting income that originate in one period and reverse in one or more subsequent periods) and the latter on temporary differences (i.e. differences between the tax base of an asset or liability and its carrying amount in the balance sheet). All timing differences are temporary differences, but there are temporary differences that do not give rise to timing differences. Because of this treatment, deferred tax assets and liabilities arose when Spanish firms adopted IFRS.

Spanish regulations establish a time limit for the recognition of deferred tax assets; the recovery of amounts that are not expected to revert within a period of 10 years is deemed uncertain, and they may not be reflected in the accounts. IAS 12 sets no time limit, and the application of IFRS therefore results in the recognition of deferred tax assets for accounting purposes.

It has not been possible to identify the issues causing the most significant quantitative impacts on accounting figures and ratios because the information provided by the sample firms was not appropriate for this purpose. It was heterogeneus and little detailed. It is a limitation of our research.

### 7. Effects of IFRS on the book-to-market ratio

The effects produced by the adoption of IFRS on the relevance of financial reporting are discussed in this section. We examine the impact of IFRS on the difference between the book and

<sup>&</sup>lt;sup>5</sup> The fair value criterion was incorporated into the Spanish rules for valuation of financial instruments in very similar terms to IAS 39 and, therefore, this matter no longer differs between the two sets of standards. The adoption of this criterion was a consequence of the need to adapt Spanish legislation to IFRS. Firms applied it for the first time in 2005.

Z statistic	Sig.
-4.076	0.000*
-4.254	0.000*
-4.457	0.000*
-4.457	0.000*
	Z statistic -4.076 -4.254 -4.457 -4.457

Table 6 Book value vs. market value (observations related to 26 firms)

\*Significant at 1%; \*\*significant at 5%.

market value of firms, which is to say the book-to-market ratio. This issue is particularly relevant because one of the reasons for the adoption of international standards was to ensure the generation of useful information for the stock market, which would imply narrowing the gap between a firm's book and market value.

#### 7.1. Definition of variables and hypothesis

We began by testing the market value of Spanish firms to establish whether the figures differ significantly from book value under Spanish standards and IFRS in both June and December. We found significant differences as shown in Table 6.

We then analyzed the gap between book and market value (measured by the book-to-market ratio—BtM) to establish whether it differed depending on the measurement of book value under local or international standards. The variables concerned are:

$$BtM_{iSAS} = \frac{B_{iSAS}}{MV_i}$$
(1)

and

$$BtM_{iIFRS} = \frac{B_{iIFRS}}{MV_i}$$
(2)

where  $BtM_{iSAS}$  is the book-to-market ratio per SAS;  $BtM_{iIFRS}$  is the book-to-market ratio per IFRS;  $B_{iSAS}$  is book value per SAS;  $B_{iIFRS}$  is book value per IFRS;  $MV_i$  is market value. The subscript *i* represents firms.

Based on these variables, we tested the null hypothesis:

**H**<sub>03</sub>. There are no significant differences in the book-to-market ratio per SAS and IFRS.

We tested the hypothesis with values at June and December.

Finally, we tested whether any evolution of market value is observable in the period analyzed (June through December 2004) in line with the evolution of book value per international standards, or whether this is closer to book value under Spanish rules.

To this end, we utilized the absolute values for relative variations in market value and book value per SAS and IFRS at June and December.

$$VAR_{iMV} = \left| \frac{MV_{idec} - MV_{ijun}}{MV_{ijun}} \right|$$
(3)

$$VAR_{iBSAS} = \left| \frac{B_{iSASdec} - B_{iSASjun}}{B_{iSASjun}} \right|$$
(4)

	Z statistic	Sig.
BtM <sub>SAShalf</sub> vs. BtM <sub>IFRShalf</sub> BtM <sub>SASyear</sub> vs. BtM <sub>IFRSyear</sub>	-3.187 -2.273	0.001* 0.023**

Table 7 Test of hypothesis  $H_{03}$  (observations related to 26 firms)

\*Significant at 1%; \*\*significant at 5%.

$$VAR_{iBIFRS} = \left| \frac{B_{iIFRSdec} - B_{iIFRSjun}}{B_{iIFRSjun}} \right|$$
(5)

The null hypotheses tested are defined as follows:

 $H_{04}$ . There are no significant differences between the relative variation of book value per SAS (VAR<sub>BSAS</sub>) and the variation in market value (VAR<sub>MV</sub>) arising in the period June through December 2004.

 $H_{05}$ . There are no significant differences between the relative variation of book value per IFRS (VAR<sub>BIFRS</sub>) and the variation in market value (VAR<sub>MV</sub>) arising in the period June through December 2004.

Descriptive statistics for the variables utilized in this section are given in Appendix C (Panel 3).

#### 7.2. Results

#### 7.2.1. Impact of IFRS on the book-to-market ratio

Having established that book value differs significantly from market value using either of the two standards, we tested  $H_{03}$  in order to discover whether differences arise in the book-to-market ratio depending on the application of SAS or IFRS and looked for any narrowing of the gap between the book and market values of the Spanish firms.

The results presented in Table 7 show that the book-to-market ratio differs significantly depending on the rules applied in both June (at 1%) and December (at 5%). Meanwhile, the sum of ranks in the Wilcoxon tests reveals that book value is actually further from market value when IFRS are applied than when Spanish accounting rules are used. Similar results were found by Niskanen et al. (2000), based on Finnish GAAP, Lin and Chen (2005), based on Chinese GAAP, and Schiebel (2006), based on German GAAP. In our survey the results are the same at both of the dates considered, although the gap appears to narrow in December.

#### 7.2.2. Evolution of market value and book value

Finally, on the basis of the tests of  $H_{04}$  and  $H_{05}$  concerning the evolution of market and book value, both hypotheses must be rejected as shown by the results presented in Table 8. Thus, the

#### Table 8

Test of hypothesis H<sub>04</sub> and H<sub>05</sub> (observations related to 26 firms)

		Z Statistic	Sig.
H <sub>04</sub>	VAR <sub>BSAS</sub> vs. VAR <sub>MV</sub>	-2.857	0.004*
H <sub>05</sub>	VAR <sub>BIFRS</sub> vs. VAR <sub>MV</sub>	-2.502	0.012**

\*Significant at 1%; \*\*significant at 5%.

relative variation in book value depending on the application of Spanish or international accounting standards, and the observed difference in market value for the period June–December 2004, are different. These differences are statistically significant (at 1% when book value is measured using Spanish accounting standards and 5% for IFRS).

This result means that the evolution of the market value of Spanish firms is not in line with their book value in the period analyzed, regardless of the rules applied by the firms to prepare their financial information. However, the different levels of signification at which hypotheses  $H_{04}$  and  $H_{05}$  are rejected, as well as the sums of ranks, indicate that the relative variation in market value differs more widely from the variation in book value measured using Spanish than international standards.

#### 8. Conclusions

This study has two objectives: (1) to establish whether the financial statements of Spanish firms are comparable when some apply IFRS and others continue to use Spanish standards and (2) to determine the effect of the adoption of IFRS on the relevance of financial reporting in Spain. We measured the quantitative impact of the application of IFRS on financial figures and ratios. We compared financial information prepared under Spanish and international accounting rules and tested for the existence or otherwise of statistically significant differences. We also analyzed the impact of IFRS on the difference between the book and market value of firms, the book-to-market ratio, to determine the relevance of IFRS.

The results of the study indicate that the image of listed Spanish firms differs significantly when IFRS rather SAS are applied in the preparation of financial information for the first half of 2004. In the balance sheet, this effect is most significant in debtors, cash and cash equivalents, equity, long-term and total liabilities. The figures are all relevant for the evaluation of a firm's financial structure.

Based on the analysis of significant adjustments made by firms to adapt their financial statements to international standards, the main causes of the significant variation in current assets were the application of fair value to financial instruments, the reclassification of accounts, and changes in the scope of consolidation. On the liability side, significant variations were due to the change in the rules for the valuation of debts and changes in the scope of consolidation. The significant variation in equity is due to direct adjustments and to the indirect effect of adjustments to results.

Fixed assets and inventories were the only items that did not vary significantly. Fixed assets did not vary because the majority of firms opted not to change the fixed asset valuation criterion applied (acquisition cost). In the case of inventories, there were no significant variations because the LIFO method, which is not permitted under IFRS, was not generally applied by Spanish firms.

In the income statement, significant variations were found in operating income depending on whether Spanish standards or IFRS were used because there are differences in the treatment of revenues and expenses (R&D expenses, asset impairment, etc.). Another difference between Spanish and international standards which had a significant impact on the firms analyzed was the treatment of extraordinary income, which made it necessary to reclassify certain extraordinary items under SAS as operating income under IFRS.

Cash, solvency and indebtedness ratios, as well as the return on assets and return on equity, varied significantly as a result of the changes in the balance sheet and income statement. We conclude that the economic and financial positions of Spanish firms, reflected in accordance with IFRS, are significantly different from the image presented by local accounting standards. Furthermore, the quantitative impact of IFRS on the interim information persists when the period analyzed is extended to the full year.

The results show that book value differs significantly from market value under both standards. The results also show that the book-to-market ratio varies significantly depending on the accounting standards applied, and is less than one in both cases. We also conclude that the gap between book and market value is wider when IFRS are applied than when Spanish standards are utilized. The conclusion is the same at both the dates considered (June and December 2004), although the difference appears to be smaller in December.

According to Feltham and Olshon (1995), accounting policies may be defined as conservative when the market value of the firm exceeds the book value (ex-ante or balance conservatism). Differences in accounting conservatism in the international context have been analyzed in studies such us Joos and Lang (1994), Arce and Mora (2002) and García and Mora (2004), which reveal that common-law countries are less conservative than code-law countries. Assuming that IFRS are nearer to common-law countries, especially the United States and United Kingdom,<sup>6</sup> the international standards may be expected to show less conservatism than SAS, because the Spanish accounting system is code-law based.

However, the manner in which Spanish firms have applied IFRS continues to provide conservative financial information. This result may be due to cultural values inherent in the Spanish accounting tradition. Consequently, it would be of interest in the future to explore changes in the book-to-market ratio in different countries following the application of IFRS in order to establish whether the pre-existing level of conservatism in local accounting standards is related to the gap between market and book value after the adoption of international accounting standards. It is likely that this conservatism, which is without doubt linked to cultural and legal factors, has some impact on the choice of recognition and valuation criteria from among the options provided by IFRS.

Based on the analysis of the evolution of market value and book value in the last quarter of 2004, we conclude that the relative variation in market value is not in line with the variation in book value, regardless of the criteria applied by the firms in the preparation of their financial information. However, the results indicate that the gap between the relative variation in market value is wider than the relative variation in book value when measured under Spanish rules rather than IFRS. Considering that the period analyzed is short and that firms have only recently begun to apply IFRS, it may be expected in the medium to long-term that international standards will result in an evolution of firms' book values that is more in line with their market value.

Our results should be of interest to the institutions involved in implementing the changes necessary to harmonize European and international accounting. Our results also may help Spanish standard setters improve the process of reforming the General Chart of Accounts in order to ensure convergence between Spanish accounting standards and IFRS for all companies. Users also should benefit from the findings because they highlight the comparability problem between firms and the absence of any improvement in relevance after the adoption of IFRS.

The study has some limitations. First, the period for the analysis is short, and it would be of interest to extend the time horizon. Likewise, the sample of firms analyzed is small. They are the firms with the highest stock market capitalization and, therefore, they are representative of the behaviour and evolution of the Spanish stock market over a given period. Nevertheless, it would be worth repeating the study for all listed companies currently applying IFRS. Finally, our study does not allow quantification of the direct effect of each standard on the accounting figures and financial ratios. This limitation would, however, be difficult to overcome, because the information furnished by firms is not sufficiently detailed and is too patchy for this purpose, as explained above.

<sup>&</sup>lt;sup>6</sup> See, for example, Rivera (1990) and Wallace (1990).

# Appendix A

Differences between IFRS and Spanish Accounting Standards<sup>a</sup>

Issue	IFRS treatment	SAS treatment	Main differences
Balance sheet areas Measurement of property, plant and equipment and intangible assets subsequent to initial recognition	IAS 16 and IAS 38: - Acquisition cost, or - fair value	GCA (Part V): - Acquisition cost	SAS do not allow fair value
Amortization of intangible assets	IAS 38: Allows for the possibility of unlimited useful life of intangible assets, which therefore need not be amortized	GCA (Part V): Does not envisage the possibility of unlimited useful life. All intangible assets are amortized	Under SAS intangible assets are always amortized. Under IFRS, assets with unlimited useful life need not be amortized
Impairment of assets	<ul> <li>IAS 36:</li> <li>Impairment loss is reflected by writing down the book value of the asset to its recoverable amount and is recognized as an expense.</li> <li>Recoverable amount is estimated for individual assets and if this is not possible it should be estimated for the cash-generating unit.</li> </ul>	GCA (Part V): - Distinguishes between reversible and irreversible impairment. Irreversible impairment (when reversal of the impairment loss is not expected) is reflected by writing down the book value of the asset to its recoverable amount and is recognized as an expense. Reversible impairment (when reversal of the impairment loss is possible) is reflected by provisions. - Recoverable amount is estimated for individual asset. Cash-generating unit concept is not included.	<ul> <li>IFRS does not distinguish between reversible and irreversible impairment. The treatment is different in the case of reversible impairment.</li> <li>SAS only envisage the estimation of recoverable amounts for individual assets but do not include the cash-generating unit concept.</li> </ul>
Measurement of inventories	IAS 2: - At the lower of cost and net realizable value. - When specific identification of costs is not possible, FIFO or weighted average cost formulas should be used.	<ul> <li>GCA (Part V):</li> <li>At the lower of cost and net realizable value.</li> <li>If specific identification of costs is not possible, FIFO, LIFO, weighted average or other cost formulas should be used.</li> </ul>	IFRS only allow the use of weighted average and FIFO cost formulas, while SAS also permit LIFO and similar methods.

167

Appendix A	(Continued)
------------	-------------

Issue	IFRS treatment	SAS treatment	Main differences
Presentation of grants related to assets	IAS 20:	GCA (Part V):	SAS do not allow to presentation of grants by deducting it in arriving at the carrying amount of the asset.
	<ul> <li>As deferred income which is recognized as income over the useful life of the asset, or</li> <li>by deducting the grant in arriving at the carrying amount of the asset. The grant is recognized as income by way of a reduced depreciation charge.</li> </ul>	As deferred income which is recognized as extraordinary income over the useful life of the asset.	When the grant is presented as deferred income, under SAS it is recognized as extraordinary income over the useful life of the asset. These amounts are not extraordinary items under IFRS.
Recognition of grants related to non-depreciable assets	IAS 20: The amount is recognized as income over the periods which bear the cost of meeting the obligations conditioning grants of this nature.	GCA (Part V): The grant is taken to income in the year the non-depreciable asset is retired or sold.	Under IFRS the grant is recognize as income to compensate the cost of certain obligations. Under SAS, the grant is not recognized as income while the asset remains in the firm.
Goodwill	IFRS 3: Goodwill is not amortized; but firm must test for impairment annually. Any impairment must be recognized.	RPCAA (Chapter III): Goodwill should be amortized systematically over the useful life, which may not exceed 20 years.	SAS establish a maximum amortization period of 20 years. Under IFRS goodwill is not amortized.
Financial Instruments	IAS 39 Financial assets: fair value (except loans and receivables, held-to-maturity investments) Financial liabilities: amortised cost.	GCA (Part V) <sup>b</sup> Financial assets: lower of acquisition cost and market. Financial liabilities: repayment value.	SAS do not allow measurement of financial assets at fair value <sup>b</sup> . Under IFRS financial liabilities should be valued at amortised cost and under SAS at repayment value.
Provisions	IAS 37: A provision should be recognized when, and only when, the firm has a present obligation, it is probable that an outflow of resources will be required to settle it, and the amount of the obligation can be reliably estimated.	GCA (Part V): Two types of provisions exist: those that correct the value of assets and those representing an obligation, which are recorded as liabilities.	IFRS only consider provisions representing an obligation, since impairments in the value of assets are recognized directly by writing down the value of the asset.

#### Income areas

Start-up costs	IAS 38: Start-up costs should be expensed when incurred.	GCA (Part V): Start-up costs are recognized as an asset when incurred and are amortized over 5 years.	Under IFRS start-up costs are expensed when incurred and under SAS they are deferred over a period of 5 years.
R&D expenditure	IAS 38: Expenditure on research should be recognize as an expense when it is incurred. Development expenditure should be recognized as an intangible asset if the firm can show the amount meets certain conditions.	GCA (Part V): Expenditure on research and development may be recognized as an intangible asset if the firm can show the amount meets certain conditions.	SAS allows the capitalization of research and development expenses, while IFRS does not permit the capitalization of research expenditure, though it requires capitalization of development expenditure when certain conditions are observed.
Borrowing costs	IAS 23: - Borrowing costs eligible for capitalization are the borrowing cost incurred less any investment income on the temporary investment of the borrowings.	GCA (Part V): - Borrowing costs eligible for capitalization are the borrowing cost incurred	SAS do not envisage any investment income on the temporary investment of the borrowings to determine borrowing costs eligible for capitalization. The limit to borrowing costs for capitalization is different: under IFRS, the amount of borrowing costs incurred, and, under SAS, the market value of the asset
	- The amount of borrowing costs capitalized during a period should not exceed the amount of borrowing costs incurred.	- The limit to borrowing costs for capitalization is the market value of the asset.	value of the asset.
Income areas			
Definition of Finance lease	IAS 17: A lease is a finance lease if it transfers substantially all risks and rewards incident to ownership. Some examples:	GCA (Part V): A lease is a finance lease when the lessee has the option to purchase the asset and there are no doubts it will happen by the	SAS confine the definition of a lease as financial to the existence of a purchase option, while IFRS refer to certain other situations

end of the lease term.

## Appendix A (Continued)

Issue	IFRS treatment	SAS treatment	Main differences
	<ul> <li>the lease transfers ownership</li> <li>the lessee has the option to purchase</li> <li>the lease term is for the major part of</li> <li>the economic life</li> <li>the present value of the minimum lease</li> <li>payments amounts to at least substantially</li> <li>all of the fair value.</li> <li>The leased asset is of specialized</li> <li>nature.</li> </ul>		
Recognition of Finance lease	IAS 17:	GCA (Part V):	Under IFRS, the liability does not include finance charges, which are included under SAS
	A finance lease should be recognized as an asset and a liability in the balance sheet at the same amount. Finance charges are recognized as an expense when incurred. The depreciation of leased assets should be over the shorter of useful life and the term of the lease in cases where subsequent ownership of the asset is not guaranteed	A finance lease should be recognized as an asset and a liability in the balance sheet. The liability includes the finance charge. Leased assets are depreciated over their useful lives.	IFRS propose depreciation of leased assets over the shorter of useful life and the term of the lease in cases where subsequent ownership of the asset is not guaranteed. Under SAS, depreciation is provided over the useful life of the asset in all cases.
Extraordinary income and expense	The annual income statement does not include income or expenses of an extraordinary nature.	GCA (Part IV): The annual income statement distinguishes between income and expenses of an ordinary and extraordinary nature. Some examples of extraordinary income (expenses) are: income (losses) from sale of property, plant and equipment, income (expense) from prior years' errors, etc.	Certain income and expenses considered extraordinary items under SAS are treated as ordinary under IFRS, which do not permit the recognition of an extraordinary result.
Income taxes	IAS 12:	GAC (Part IV)	IFRS require for the treatment of deferred taxes using the balance sheet liability method, while GAC requires the income statement liability method.
	This requires an entity to account for deferred tax using the balance sheet liability method.	The rule requires an entity to account for deferred tax using the income statement liability method.	Under IFRS, an entity must offset deferred tax assets and deferred tax liabilities, which is prohibited under SAS.

A deferred tax asset is recognized when it is probable that taxable income will be available against which the deductible temporary difference can be utilized (no temporal limit).

An entity shall offset deferred tax assets and deferred tax liabilities of the same taxable entity if it complies with certain conditions.

Foreign currency transactions are recorded, on initial recognition, in the functional currency. Exchange differences arising at each balance sheet date are recognized in profit or loss in the period in which they arise.

#### Investment in associates

Other areas Foreign currency

IAS 28:

IAS 21:

An associate is an enterprise in which the investor has significant influence and which is neither a subsidiary nor a joint venture of the investor. It is presumed that the investor has significant influence when the investor holds 20% or more of the voting power of the investee.

#### A deferred tax asset is recognized when future recovery is likely (within a maximum period of 10 years).

An entity cannot offset deferred tax assets and deferred tax liabilities.

#### RPCAA (Chapter V) GCA (Part V):

A foreign currency transaction is recorded, on initial recognition, in the presentation currency. At each balance sheet date, exchange gains are, in general, deferred until realized. Negative exchange differences are recognized as an expense in the period in which they arise. SAS also allow the capitalization of exchange differences associated with liabilities incurred in the acquisition or construction of assets under certain circumstances.

#### RPCAA (Chapter I):

An associate is an enterprise in which the investor has significant influence and which is neither a subsidiary nor a joint venture of the investor. It is presumed that the investor has significant influence when the investor holds 20% or more (3% if the firm is listed on a stock market) of the voting power of the investee.

Under IFRS, exchange gains or losses arising at the balance sheet date must be taken to results. Under SAS, the treatment of negative exchange differences is the same, but it differs for exchange gains. Exchange gains are, in general, deferred until realized. The capitalization of exchange differences associated with liabilities incurred in the acquisition or construction of assets under certain circumstances is another difference between IFRS and SAS Under both standards an investee company should be treated as an associate in the case of significant influence. However IFRS establish that such influence exists when the interest held is equal to or greater than 20%, while SAS establish a limit of 3% if the investee company is listed.

Appendix A (Continued)

Issue	IFRS treatment	SAS treatment	Main differences
Recognition of changes in accounting policies	IAS 8:	GCA (Part V):	IFRS require changes in accounting policies to be recognized
	Changes in accounting policies should be recognized retrospectively. The entity is required to adjust the opening balance of each affected component of equity for the earliest prior period presented and the comparative amounts, as if the new accounting policy had always been applied	Changes in accounting policies causes an adjustment, which is recognized as extraordinary income or expense for the year in which the change is made. Comparative amounts are not corrected.	retrospectively, which is to say as if the new criteria had always been applied. Under SAS, these adjustments are recognized as extraordinary results for the year in which the change is made, and no correction is made to the comparative information
Recognition of errors	IAS 8: The financial statements and the comparative amounts should be presented as if the error had been corrected when it occurred.	GCA (Part V): A prior period error causes an adjustment which should be recognized as extraordinary income or expense for the year in which the error is corrected.	In accordance with IFRS, the financial statements (including comparative statements) must be presented as if the error had been corrected in the period in which it occurred. This criterion is not followed by SAS, which require adjustments to be recognized as extraordinary results.

SAS: Spanish accounting standards; GCA: General Chart of Accounts; RPCAA: Regulation for preparation of consolidated annual accounts; IAS: International accounting standards; IFRS: International Financial Reporting Standards.

<sup>a</sup> Some issues regulated in IFRS are not regulated in depth in SAS. Because of this, such issues could also result in differences in the financial statements when they are prepared under IFRS rather than SAS. Some examples are: investment property, non-current assets held for sale, financial instruments, employee benefits, financial reporting in hyperinflationary economies, etc.

<sup>b</sup> The fair value criterion was incorporated into the Spanish rules for valuation of financial instruments in very similar terms to IAS 39, and this matter therefore no longer differs between the two sets of standards. The adoption of this criterion was a consequence of the need to adapt Spanish legislation to IFRS. Firms applied it for the first time in 2005.

172

	Kolmogorov–Smirnov		Shapiro–Wilk	
	Statistic	Sig.	Statistic	Sig.
Panel 1. Normality test for ha	alf-vear accountir	g figures and financial ratios	(observations related to 26 fi	rms)
Fixed assetss A shalf	0.266	0.000	0.670	0.000
Inventoriess	0.184	0.019	0.827	0.000
DebtorssAshalf	0.187	0.016	0.832	0.001
Cashs A shalf	0.202	0.006	0.805	0.000
Current assetss A shalf	0.196	0.009	0.854	0.001
Total assetss A shalf	0.239	0.000	0.713	0.000
Equitys A shalf	0.285	0.000	0.651	0.000
L.T. liabilitiess Ashalf	0.272	0.000	0.671	0.000
S T. liabilitiess A shalf	0.206	0.005	0.818	0.000
Total liabilitiess A shalf	0.221	0.002	0.730	0.000
L.T. resourcess A shalf	0.262	0.000	0.675	0.000
Equity + liabilitiess $A$ shalf	0.239	0.000	0.713	0.000
OPIs A shalf	0.361	0.000	0.645	0.000
ORDISAShalf	0.356	0.000	0.656	0.000
NETISAShali	0 359	0.000	0.712	0.000
ATTISAShalf	0.341	0.000	0.719	0.000
Current ratios Ashale <sup>a</sup>	0.095	0.200	0.966	0.507
Acid tests a shale <sup>a</sup>	0.112	0.200	0.950	0.216
Cash rationashak	0.183	0.021	0.780	0.000
Solvencycashalf	0.219	0.002	0.883	0.006
Indebtednesser or ac	0.227	0.002	0.804	0.000
ROA (OPI)s A shalf	0.203	0.006	0.811	0.000
ROA (ORDI) at all 16	0.197	0.009	0.775	0.000
ROF (ORDI) SAShalf	0.253	0.000	0.805	0.000
ROE (NETI) SAShalf	0.283	0.000	0.755	0.000
Fixed assets	0.257	0.000	0.671	0.000
Inventories	0.198	0.008	0.806	0.000
Debtorsupper	0.203	0.006	0.803	0.000
Cashurper us	0.205	0.000	0.305	0.000
Current assetsurpor is	0.200	0.000	0.853	0.000
Total assetsurper as	0.242	0.007	0.855	0.001
Fourtywork is	0.242	0.000	0.687	0.000
L T liabilities	0.292	0.000	0.661	0.000
S T liabilities	0.204	0.005	0.781	0.000
Total liabilities	0.229	0.005	0.718	0.000
I T resources	0.223	0.000	0.667	0.000
E. I. Tesources <sub>IFRShalf</sub>	0.233	0.000	0.714	0.000
	0.350	0.000	0.670	0.000
ORDIERGE	0.335	0.000	0.679	0.000
NETLER at the	0.333	0.000	0.686	0.000
	0.314	0.000	0.687	0.000
Current rationspanes <sup>a</sup>	0.106	0.000	0.007	0.000
Acid testurner w <sup>a</sup>	0.145	0.200	0.973	0.074
Cash rationration	0.175	0.000	0.938	0.337
Solvencyurner	0.181	0.000	0.853	0.000
Indebtednessurpar	0.101	0.023	0.865	0.001
ROA (OPI)	0.222	0.002	0.817	0.002
ROA (ORDI)TED Shalf	0.221	0.002	0.802	0.000

# Appendix B. Normality test

# Appendix B (Continued)

	Kolmogorov–Smirnov		Shapiro–Wilk	
	Statistic	Sig.	Statistic	Sig.
ROE (ORDI)IFRShalf	0.228	0.001	0.841	0.001
ROE (NETI) <sub>IFRShalf</sub>	0.218	0.002	0.832	0.001
Panel 2. Normality test for ar	nual accounting fig	ures and financial ratios (o	bservations related to 26 firm	s)
Fixed assetss A super	0.272	0.000	691	.000
Inventoriess A sugar	0.206	0.005	823	.000
Debtorse A sugar	0.191	0.013	838	001
Cashs A sugar	0.329	0.000	591	.000
Current assetse A sugar	0.211	0.003	854	001
Total assetss A super	0.237	0.000	735	.000
Fauitys A sugar	0.269	0.000	663	000
I T liabilities	0.265	0.000	709	000
S T liabilities A Swar	0.205	0.000	767	000
Total liabilitieSata	0.214	0.003	751	000
I T resources	0.258	0.002	701	000
E. I. IesourcessAsyear	0.238	0.000	725	.000
Current rotio	0.237	0.000	.735	.000
A aid test	0.105	0.200	.9/1	.022
Coch ratio	0.174	0.050	.890	.011
Cash ratio <sub>SASyear</sub>	0.340	0.000	.500	.000
Solvency <sub>SASyear</sub>	0.263	0.000	.852	.001
Indebtedness <sub>SASyear</sub>	0.237	0.000	.688	.000
Fixed assets <sub>IFRSyear</sub>	0.279	0.000	.693	.000
Inventories <sub>IFRSyear</sub>	0.211	0.003	.799	.000
Debtors <sub>IFRSyear</sub>	0.190	0.013	.818	.000
Cash <sub>IFRSyear</sub>	0.279	0.000	.6/2	.000
Current assets <sub>IFRSyear</sub>	0.217	0.002	.850	.001
Total assets <sub>IFRSyear</sub>	0.229	0.001	.736	.000
Equity <sub>IFRSyear</sub>	0.268	0.000	.708	.000
L.T. liabilities <sub>IFRSyear</sub>	0.306	0.000	.681	.000
S.T. liabilities <sub>IFRSyear</sub>	0.213	0.003	.750	.000
Total liabilities <sub>IFRSyear</sub>	0.225	0.001	.736	.000
L.T. resources <sub>IFRSyear</sub>	0.256	0.000	.690	.000
Equity + liabilities <sub>IFRSyear</sub>	0.229	0.001	.736	.000
Current ratio <sub>IFRSyear</sub> <sup>a</sup>	0.081	0.200	.974	.709
Acid test <sub>IFRSyear</sub> <sup>a</sup>	0.129	0.200	.942	.136
Cash ratio <sub>IFRSyear</sub>	0.252	0.000	.694	.000
Solvency <sub>IFRSyear</sub>	0.230	0.001	.839	.001
IndebtednessIFRSyear	0.234	0.001	.868	.003
Panel 3. Normality test for va	riables related to be	ook and market values (obs	ervations related to 26 firms)	
MV <sub>half</sub>	0.340	0.000	0.499	0.000
MVyear	0.275	0.000	0.594	0.000
BtMsAshalf	0.287	0.000	0.611	0.000
BtMIERsholf	0.258	0.000	0.669	0.000
BtMsAsvear	0.253	0.000	0.650	0.000
BtMIERSvear	0.214	0.003	0.768	0.000
VAR <sub>MV</sub>	0.379	0.000	0.442	0.000
VARDEAE	0.187	0.020	0.843	0.000
VAR	0.218	0.003	0.750	0.001
****BIFRS	0.210	0.005	0.750	0.000

<sup>a</sup> Variables found to be normal.

175

	Min.	Max.	Mean	St. dev.
Panel 1. Half-year accounting	g figures (thousands	of euros) and financial r	atios. Observations relat	ed to 26 firms
Fixed assets <sub>SAShalf</sub>	191069	48807343	8564971	12186920
InventorieS <sub>SAShalf</sub>	182	2423000	613723	667286
Debtors <sub>SAShalf</sub>	206992	9290449	2423968	2424353
Cash <sub>SAShalf</sub>	1772	639990	140090	159094
Current ratio <sub>SAShalf</sub> <sup>a</sup>	304473	10530236	3177780	2911118
Acid test <sub>SAShalf</sub> <sup>a</sup>	638427	59337579	11742752	14512812
Equity <sub>SAShalf</sub>	145941	19162683	4030176	5515859
L.T. liabilities <sub>SAShalf</sub>	60286	25756350	4388105	6227689
S.T. liabilities <sub>SAShalf</sub>	212833	14237749	3293780	3401346
Total liabilities <sub>SAShalf</sub>	273119	39994099	7681885	9341511
L.T. resourcessAshalf	293006	44919033	8418281	11403499
Equity + liabilitiessAshalf	622679	59156782	11712061	14487326
OPIsashalf	-42952	3420954	555696	804732
ORDIsashalf	-102958	2696319	455996	650357
NETISAShalf	-89964	1411054	302710	383823
ATTISAShalf	-81952	1254177	276446	344066
Current ratios Ashalf	0.3700	1 9906	1.1110	0 4005
Acid tests Ashalf	0.3617	1 5435	0.8505	0.3128
Cash ratios A shalf	0.0028	0 2064	0.0450	0.0427
SolvencysAshah	1 1072	2 6054	1 6040	0.3938
Indebtednesse A shalf	0.6229	9 3262	2 6118	2 0130
ROA (OPI) s A Shalf	-0.0176	0.2310	0.0531	0.0450
ROA (ORDI)sashali	-0.0421	0.2343	0.0459	0.0466
ROF (ORDI) SAShalf	-0.4352	0.4954	0.1297	0.1646
ROE (NETI) s A SH AS	-0.3803	0.3553	0.0889	0.1246
Fixed assets	185907	46257898	8394736	11844466
Inventories	182	2570820	650949	737811
Debtorsympos	188375	0588608	216/385	2307023
Cashurper	3472	2052000	36/113	481496
Current assetsurnal is	2000/5	10010260	3170447	2030311
Total assetsurp at 16	638507	57168167	1157/18/	14266645
Fouitvrenau as	122507	13171000	31/802/	3874412
I T liabilities	59960	29951657	5177272	76/1889
S T liabilities man	238380	15375350	32/7088	3420861
Total liabilities	298340	45327016	8425260	10688006
I T resourcesurper as	270340	41702808	8326106	11301627
Equity   liabilitiogare as a	628507	57162167	1157/19/	14266645
OPIver a va	46857	2874476	523104	74200043
OPDI	118200	2074470	142624	615102
NETI	-118200	1582647	200242	424550
ATTI	-00137	1363047	202042	206241
Al HIFRShalf	-80123	1404002	292042	0 291/
Tatal assets	0.3703	1.9901	0.8220	0.3614
Cooh rotio	0.018/	1.4445	0.8220	0.5148
Cash ratio <sub>IFRShalf</sub>	0.0055	0.5706	0.1348	0.14/7
Solvency <sub>IFRShalf</sub>	1.1549	2.2845	1.5025	0.3410
Indebtedness <sub>IFRShalf</sub>	0.7785	6.4563	3.0254	1.9212
KOA (OPI) <sub>IFRShalf</sub>	-0.01//	0.2166	0.0496	0.0426
KOA (UKDI)IFRShalf	-0.0446	0.2189	0.0448	0.0444
KOE (OKDI) <sub>IFRShalf</sub>	-0.3326	0.480/	0.1462	0.1516
ROE (NETI) <sub>IFRShalf</sub>	-0.2480	0.3168	0.1016	0.1042

Appendix C. Descriptive statistics

# Appendix C (Continued)

	Min.	Max.	Mean	St. Desv.
Panel 2. Annual accounting figu	res (thousands of eu	ros) and financial ratios	. Observations related to	o 26 firms
Fixed assets <sub>SASyear</sub>	189689	50982288	9140493	12665136
Inventoriess	172	2652000	668457	741069
Debtors	187740	9427797	2446015	2419533
Cashs A syear	300	3303000	453787	805443
Current ratios A sugar <sup>a</sup>	303801	12126000	3568259	3338302
Total assetss Asyear	627702	61934715	12708752	15338040
Equitys Asyear	180895	19230305	4254394	5633932
L.T. liabilitiess Asympt	66944	23150655	4572989	6094337
S.T. liabilitiess A Syear	201775	19553755	3881370	4428568
Total liabilitiess Asvear	268719	42704410	8454358	10097295
L.T. resourcessasyear	346689	42380960	8827382	11411097
Equity + liabilitiess A sugar	627702	61934715	12708752	15338040
Current ratio	0.3267	2.1638	1.1549	0.4654
Acid testingsycar <sup>a</sup>	0.3215	2,1629	0.8879	0.4035
Cash ratios A svear	0.0005	1.2325	0.1388	0.2547
Solvencys A syear	1.0841	2.7277	1.6123	0.4192
Indebtednesss	0.5788	11.8917	2,6337	2.3666
Fixed assets	188810	48932285	9063921	12447570
Inventoriesiepsvear	172	2901075	697619	807833
Debtorsiersvear	107195	9576706	2293022	2420879
CashiFR Syear	2027	3321000	541601	807375
Current assets <sub>IFR Syear</sub>	301936	11966000	3532241	3320751
Total assets	630466	60078856	12596163	15182723
Equity <sub>IERSyear</sub>	208314	14033000	3456881	4071896
L.T. liabilities <sub>IFRSyear</sub>	66601	27458526	5336712	7601844
S.T. liabilities	208482	19993802	3802569	4400444
Total liabilities <sub>IFRSyear</sub>	275083	47452328	9139281	11438136
L.T. resources <sub>IFRSvear</sub>	331161	40538000	8793594	11464859
Equity + liabilities <sub>IFRSyear</sub>	630466	60078856	12596163	15182723
Current ratio <sub>IFRSyear</sub>	0.3315	2.0817	1.1385	0.4397
Acid test <sub>IFRSyear</sub>	0.1812	2.0808	0.8577	0.4102
Cash ratio <sub>IFRSvear</sub>	0.0030	1.1928	0.1873	0.2590
Solvency <sub>IFRSvear</sub>	1.1329	2.7383	1.5481	0.3930
Indebtedness <sub>IFRSyear</sub>	0.5753	7.5238	2.8195	1.8885
Panel 3. Variables related to boo	k and market values	. Observations related t	o 26 firms	
B <sub>SAShalf</sub> (thousands of euros)	145941	19162683	4030176	5515859
B <sub>IFRShalf</sub> (thousands of euros)	122507	13171000	3148924	3874412
$MV_{half}$ (thousands of euros)	887785	61442939	7258722	12228248
BtMsAshalf	0.0566	2.3234	0.6503	0.4625
BtMIFRShalf	0.0610	2.0366	0.5599	0.4021
B <sub>SASyear</sub> (thousands of euros)	180895	19230305	4254394	5633932
B <sub>IFRSyear</sub> (thousands of euros)	) 208314	14033000	3456881	4071896
MV <sub>vear</sub> (thousands of euros)	1170227	69432038	10384986	14848645
BtM <sub>SASyear</sub>	0.0431	0.9582	0.4500	0.2790
BtM <sub>IFRSyear</sub>	0.0698	0.7760	0.4036	0.2399
VAR <sub>MV</sub>	0.0039	9.9207	0.9370	2.1494
VAR <sub>BSAS</sub>	0.0000	0.5800	0.1404	0.1352
VAR <sub>BIFRS</sub>	0.0200	0.7000	0.1610	0.1543

<sup>a</sup> Variables found to be normal.

#### References

- Alford, A., Jones, J., Leftwich, R., & Zmikewski, M. (1993). The relative informativeness of accounting disclosures in different countries. *Journal of Accounting Research*, 31(Suppl.), 183–223.
- Amir, E., Harris, T. S., & VenutiI, E. K. (1993). A Comparison of the value-relevance of U:S: versus Non-U.S. GAAP accounting measures using form 20-F Reconciliations. *Journal of Accounting Research*, 31(Suppl.), 230–264.
- Arce, M., & Mora, A. (2002). Empirical evidence of the effect of European accounting differences on the stock market valuation of earnings and book value. *The European Accounting Review*, 11, 573–599.
- Bandyopadhyay, S. P., Hanna, J. D., & Richardson, G. (1994). Capital Market effects of U.S.–Canada GAAP differences. Journal of Accounting Research, 32(2), 262–277.
- Bartov, E., Goldberg, S. R., & Kim, M. (2005). Comparative value relevance among German, US and International Accounting Standards: A German stock market perspective. *Journal of Accounting Auditing & Finance*, 20(2), 95– 119.
- Callao, S., Jarne, J. I., & Laínez, J. A. (2006). ¿Cómo han tratado los grupos cotizados españoles el proceso de adopción de las NIC/NIIF? Observatorio Contable y Financiero, 1, 32–49.
- Chan, K. C., & Seow, G. S. (1996). The association between stock returns and foreign GAAP earnings versus earnings adjusted to U.S. GAAP. *Journal of Accounting and Economics*, 21, 139–158.
- Circular 1/2005, April 1, of National Securities Market Commission, to modify the periodical public information formats for listed companies. *Official Bulletin* No. 82, 6 April 2005.
- Delvaille, P., Ebbers, G., & Saccon, C. (2005). International financial reporting convergence: evidence from three continental European countries. Accounting in Europe, 2, 137–164.
- European Union 1978. Directive 78/660/EEC of the European Parliament and of the Council of 25 July 1978 on the annual accounts of certain types of companies. Official Journal L 222, 14 August 1978.
- European Union 1983. Directive 83/349/EEC of the European Parliament and of the Council of 13 June 1983 on the consolidated accounts. Official Journal L 193, 18 July 1983.
- Feltham, G., & Olshon, J. A. (1995). Valuation and clear surplus accounting for operating and financial activities. Contemporary Accounting Research, 11, 689–731.
- García, J. M., & Mora, A. (2004). Balance sheet versus earnings conservatism in Europe. The European Accounting Review, 13, 261–292.
- Harris, M. S., & Muller, K. A., III. (1999). The market valuation of IAS versus US-GAAP accounting measures using Form 20-F reconciliations. *Journal of Accounting & Economics*, 26(1–3), 285–312.
- Harris, T. S., Lang, M., & Möller, H. P. (1994). The value relevance of German accounting measures: An empirical analysis. *Journal of Accounting Research*, 32(2), 187–209.
- Horton, J., & Serafeim, G. (2006). Market response to and the value relevance of reconciliation adjustments from UK GAAP to IFRS GAAP: First evidence from the UK. Available at SSRN: http://ssrn.com/abstract=923582.
- Jermakowicz, E. K. (2004). Effects of adoption of international financial reporting standards in Belgium: The evidence from BEL-20 companies. Accounting in Europe, 1, 51–70.
- Jones, T. C., & Luther, R. (2005). Anticipating the impact of IFRS on the management of German manufacturing companies: Some observations from a British perspective. Accounting in Europe, 2, 165–194.
- Joos, P., & Lang, M. (1994). The effects of accounting diversity: Evidence from the European Union. *Journal of Accounting Research*, 32(Suppl.), 141–168.
- Laínez, J. A., Callao, S., & Jarne, J. I. (2004). Adopción por primera vez de las Normas Internacionales de Información Financiera. Monograph on IFRS. AECA, 5, 493–598.
- Larson, R. K., & Street, D. L. (2004). Convergence with IFRS in an expanding Europe: Progress and obstacles identified by large accounting firms' survey. *Journal of International Accounting, Auditing and Taxation*, 13, 89–119.

Law 62/2003, 30 December, on fiscal, administrative and social measure. Official Bulletin No. 313, 31 December 2003.

- Lin, Z. J., & Chen, F. (2005). Value relevance of international accounting standards harmonization: Evidence from A-share and B-share markets in China. Journal of International Accounting, Auditing and Taxation, 14(2), 79–103.
- Mora, A., & Rodríguez, A. (2004). Las Empresas de la Europa continental en NYSE: La relevancia de la conciuliación del 20-F. Revista Española de Financiación y Contabilidad, 33, 1103–1137.
- Niskanen, J., Kinnunen, J., & Kasanen, E. (2000). The value relevance of IAS reconciliation: Empirical evidence from Finland. *Journal of Accounting and Public Policy*, 19(2), 119–137.
- Ormrod, P., & Taylor, P. (2004). The Impact of the change to International Accounting Standards on Debt Covenants: A UK perspective. Accounting in Europe, 1, 71–94.
- Regulations for the Preparation of Consolidated Annual Accounts 1991. Royal Decree 1815/1991, 20 December, which approves the Regulations for the Preparation of Consolidated Annual Accounts.

- Rivera, J. (1990). International accounting standards harmonization and the role of IASC. Paper presented to the XIII Annual Congress of the European Accounting Association.
- Schiebel, A. (2006). Value relevance of German GAAP and IFRS consolidated Financial Reporting: An empirical analysis on the Frankfurt Stock Exchange. Available at SSRN: http://ssrn.com/abstract=916103.
- Schipper, K. (2005). The introduction of International Accounting Standards in Europe: Implications for international convergence. *European Accounting Review*, 14(1), 101–126.
- Spanish General Chart of Accounts 1990. Royal Decree 1643/1990, 20 December, which approves the General Chart of Accounts.
- Tokar, M. (2005). Convergence and the implementation of a single set of global standards: the real-life challenge. *Accounting in Europe*, 2, 47–68.
- Van Tendeloo, B., & Vanstraelen, A. (2005). Earnings management under German GAAP versus IFRS. European Accounting Review, 14(1), 101–126.
- Wallace, R. (1990). Survival strategies of a global organization: the case of the International Accounting Standard Committee. *Accounting Horizons*, *4*, 1–22.
- Weißenberger, B. E., Stahl, A. B., & Vorstius, S. (2004). Changing from German GAAP to IFRS or US GAAP: A survey of German companies. Accounting in Europe, 1, 169–189.